Author's response to reviews

Title: Chronic virus infections suppress atopy but not asthma in a set of children from a large Latin American city: A cross-section study

Authors:

Rafael V Veiga (rafael.veiga@uol.com.br)
Sérgio S Cunha (cunhatmt@hotmail.com)
Vitor CC Dattoli (vitor.dattoli@gmail.com)
Álvaro C Cruz (cruz.proar@gmail.com)
Philip J Cooper (pcooper@ecnet.ec)
Laura C Rodrigues (laura.Rodrigues@lshtm.ac.uk)
Maurício L Barreto (mauricio@ufba.br)
Neusa M Alcantara-Neves (neuza@ufba.br)

Version: 7 Date: 17 March 2011

Author's response to reviews:

We correct the title in the submission page.
We corrected the tables according to the editor suggestions.
We corrected minor spelling errors.
We deleted the cover letter from the submission as suggested by the Editor